Author's response to reviews

Title: Spatial analysis of factors associated with HIV infection among young people in Uganda, 2011.

Authors:

Lucy A Chimoyi (chimoyi@hotmail.com)
Eustasius Musenge (Eustasius.Musenge@wits.ac.za)

Version: 2 Date: 23 May 2014

Author's response to reviews: see over
To the Editor

BMC Public Health

Date: 23 May 2014

Dear Ms. Rita Aguirre,

RE: Manuscript Re-submission

We would like to resubmit our manuscript entitled “Spatial analysis of factors associated with HIV infection among young people in Uganda, 2011” to your journal for possible publication. This is after addressing the comments and suggested by the reviewers.

Attached are point to point responses to the reviewers' comments for you review.

We look forward to a positive response from you.

Yours Sincerely,

Lucy Chimoyi